Author’s response to reviews

Title: A framework to assess welfare mix and service provision models in health and social care: theory and practice in two prominent Italian regions

Authors:

Francesco F Longo (francesco.longo@unibocconi.it)
Elisabetta E Notarnicola (elisabetta.notarnicola@unibocconi.it)
Stefano S Tasselli (stefano.tasselli@unibocconi.it)

Version: 3
Date: 10 June 2014

Author’s response to reviews:

Dear Editor,

Following your email that we received today, the 10th of June 2014, we provide a revised version of our paper “A framework to assess welfare mix and service provision models in health and social care: theory and practice in two prominent Italian regions” to follow your suggestions and provide all the changes required.

1) We add an initial title page including full names, institutions, countries and email addresses of all authors and the full postal address of the submitting author.

2) We change our abstract to fit the four sections you suggested: Background, Methods, Results, Conclusions.

3) Our research do not involve human subjects, human material, or human data but we have add a final section attesting ethical adherence. We also attest for the absence of competing interests.

4) We add a section describing the authors contributions.

5) We include tables, that we originally presented in a separated file, at the end of the manuscript, after the references.

Hoping that we satisfied your requirements, We thank you for your attention and we are willing to answer any other questions you might have.

Best regards,

Francesco Longo
Elisabetta Notarnicola
Stefano Tasselli

CERGAS - Health and Social Care Management Research Center,
Public Policy Analysis and Management Department BOCCONI UNIVERSITY
Via Roentgen 1, MILAN, ITALY